Author’s response to reviews

Title: Consistency between stated and revealed preferences: a Discrete Choice Experiment and a behavioral experiment on vaccination behavior compared.

Authors:

Mattij S Lambooij (mattijs.lambooij@rivm.nl)
Irene A Harmsen (irene.harmsen@rivm.nl)
Jorien Veldwijk (jorien.veldwijk@rivm.nl)
Hester de Melker (hestergde.melker@rivm.nl)
Liesbeth Mollema (liesbeth.mollema@rivm.nl)
Yolanda WM van Weert (Yolanda.van.weert@rivm.nl)
G Ardine de Wit (ardine.de.wit@rivm.nl)

Version: 8 Date: 9 February 2015

Author’s response to reviews: see over
Changes MS 5202296831080017.
Request email 06-02-2015

Please re-format their title page according to the suggestions below.

- email of all authors should be included in the first page

We have added the email addresses of all authors on the title page after each author name

- Keywords should be removed from below the title, together with "Number figures: 0
  Number tables: 4" and "Running Title: Stated versus revealed preferences"

We have removed
  - the keywords;
  - the text "Number figures: 0 Number tables: 4" and ;
  - the text "Running Title: Stated versus revealed preferences".